Author's response to reviews

Title: A retrospective follow up study on maternal age and infant mortality in two Sicilian districts

Authors:

Walter Mazzucco (waltermazzucco@unipa.it)
Rosanna Cusimano (rosanna.cusimano@libero.it)
Maurizio Macaluso (macaluso@comcast.net)
Claudio La Scola (clasc@libero.it)
Giovanna Fiumanò (achille.cernigliaro@regione.sicilia.it)
Salvatore Scondotto (salvatore.scondotto@regione.sicilia.it)
Achille Cernigliaro (achille_c@libero.it)
Giovanni Corsello (giocors@alice.it)
Giuseppe La Torre (giuseppe.latorre@rm.unicatt.it)
Francesco Vitale (francesco.vitale@unipa.it)

Version: 8 Date: 6 October 2011

Author's response to reviews:

To Journal Editor
BMC Public Health

In agreement with the co-authors, we have revised the manuscript according to "Formatting changes to be made".

Best regards.
Walter Mazzucco

Major revisions:

1. References: Please provide functioning link to http://www.istat.it/it/archivio/mortalità

We have changed the old link with the following one:
http://www.istat.it/it/archivio/14562

Minor revisions:

1. Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

We have uploaded a ppt version of the figure.